Author's response to reviews

**Title:** Immunohistochemical Analysis of Oxidative Stress and DNA Repair Proteins in Normal Mammary Tissue and Breast Cancer Tumors

**Authors:**

Carol D Curtis (ccurtis@illinois.edu)  
Daniel L Thorngren (dthorng2@illinois.edu)  
Ann M Nardulli (anardull@life.uiuc.edu)

**Version:** 5 **Date:** 15 December 2009

**Author's response to reviews:** see over
Dear Sabina,

We were very pleased to learn that our manuscript was acceptable for publication in BMC Cancer. In response to Reviewer 2’s comment, we revised a section of the manuscript and figure legend 1. In order to meet formatting requirements, we have modified the Authors’ contribution section, removed the table from the figure section and included it in the manuscript, corrected the title, included the country affiliation of the authors, moved the acknowledgements, deleted the disclosure statement, renamed sections appropriately, added information on reference 17, removed the figure labels, and checked the manuscript for errors. Please let me know if any additional modifications are needed.

Thanks for your assistance,
Ann